Author's response to reviews

Title: Localized gastric amyloidosis differentiated histologically from scirrhous gastric cancer using endoscopic mucosal resection: a case report

Authors:

Tsugumasa Kamata (tkamata@ncc.go.jp)
Haruhisa Suzuki (harusuzu@ncc.go.jp)
Shigetaka Yoshinaga (shiyoshi@ncc.go.jp)
Satoru Nonaka (snonaka@ncc.go.jp)
Takeo Fukagawa (tfukagaw@ncc.go.jp)
Hitoshi Katai (hkatai@ncc.go.jp)
Hirokazu Taniguchi (hitanigu@ncc.go.jp)
Ryoji Kushima (rkushima@ncc.go.jp)
Ichiro Oda (ioda@ncc.go.jp)

Version: 6 Date: 15 April 2012

Author's response to reviews: see over
April 15, 2012

Dear Journal of Medical Case Reports Editor

Manuscript ID: 5165804676650759
Manuscript Title: Localized gastric amyloidosis differentiated histologically from scirrhou gastric cancer using endoscopic mucosal resection

Point-by-Point Responses To Reviewer Comments and Manuscript Revisions

Thank you for your letter and the comments from the reviewers concerning our manuscript entitled “Localized gastric amyloidosis differentiated histologically from scirrhous gastric cancer using endoscopic mucosal resection”.

After careful consideration, we’ve revised the manuscript and responded to those comments on a point-by-point basis. It should also be mentioned that a native-English editor has reviewed all the materials and made a number of changes to improve the overall clarity, consistency and readability of our revised manuscript.

We appreciate your further consideration and trust that the attached revised manuscript will be accepted for publication in Journal of Medical Case Reports.

Sincerely yours,

Haruhisa Suzuki, MD
Corresponding Author
Endoscopy Division
National Cancer Center Hospital
5-1-1 Tsukiji, Chuo-ku, Tokyo 104-0045, Japan
Tel: +81-3-3542-2511, Fax: +81-3-3542-3815
E-mail: harusuzu@ncc.go.jp
**Reviewers' Comments to Author:**

**Q1.** Please include the ethnicity of the patient **BOTH** in the abstract and Case presentation section.

**A1.** We agree with the reviewer’s comment and we converted the phrase: at page 2 line 7 and page 4 line 8.